Author's response to reviews

Title: Older age does not influence CD4 cell recovery in HIV-1 infected patients receiving Highly Active Anti Retroviral Therapy

Authors:

Mario M Tumbarello (tumbarello@rm.unicatt.it)
Ricardo R Rabagliati (rabagli@med.puc.cl)
Katleen K de Gaetano Donati (kdegaetanodonati@yahoo.com)
Silvia S Bertagnolio (bertagnolios@who.int)
Eva E Montuori (eva.montuori@libero.it)
Enrica E Tamburrini (enricatamburrini@rm.unicatt.it)
Evelina E Tacconelli (etaccone@caregroup.harvard.edu)
Roberto R Cauda (rcauda@rm.unicatt.it)

Version: 5 Date: 27 October 2004

Author's response to reviews: see over
Dear Editor,

This communication will serve as our response to the critical comments regarding the manuscript entitled “Older age does not influence CD4 cell recovery in HIV-1 infected patients receiving Highly Active Antiretroviral Therapy” (Manuscript 1288593109290465), submitted to *BMC Infectious Diseases*. As requested we have modified the manuscript according to the specific formatting checklist.

In closing, on behalf of my colleagues and myself, I would like to express our appreciation for the editor’s critical comments and suggestions for this manuscript and very much hope that, in the present format, our scientific effort fulfills the criteria for publication in *BMC Infectious Diseases*.

Sincerely Yours

Mario Tumbarello M.D.
Istituto Clinica Malattie Infettive
Università Cattolica Sacro Cuore
Largo Gemelli 8
00168 Roma - Italy

FAX +39-06-3054519
e-mail: tumbarello@rm.unicatt.it